Author's response to reviews

Title: Effect of meteorological variables on the incidence of hand, foot, and mouth disease in children: a time-series analysis in Guangzhou, China

Authors:

Yong Huang (hyang5556@163.com)
Te Deng (dengte2008@126.com)
Shicheng Yu (shicheng_yu@hotmail.com)
Jing Gu (gujing5@mail.sysu.edu.cn)
Cunrui Huang (huangcunrui@hotmail.com)
Gexin Xiao (biocomputer@126.com)
Yuantao Hao (haoyt@mail.sysu.edu.cn)

Version: 5 Date: 1 March 2013

Author's response to reviews: see over
March 1, 2013

Dear Editor,

Re: 1530366807826672 - Effects of meteorological variables on the incidence of hand, foot, and mouth disease in children: a time-series analysis in Guangzhou, China

Many thanks for your email dated on March 1, 2013, and attached comments. Here we have now revised our manuscript according to the editor’s comment. A point-to-point response to the editor’s comment is attached below this letter.

We are very grateful for your consideration of publishing our manuscript in *BMC Infectious Diseases*.

Yours sincerely,

Yuantao Hao
Corresponding author

**EDITOR'S COMMENTS:**

1. “With respect to the response to Q 7, however, I would suggest that 'individual information accuracy' be replaced with 'accuracy of the information reported' for clarification prior to publication.”

   Changed as suggested.